Investigating the Relationship between Cerebral Blood Flow and Cognitive Function in Hemodialysis Patients

Mark Duncan Findlay,^{1,2} Jesse Dawson,¹ David Alexander Dickie,¹ Kirsten P. Forbes,³ Deborah McGlynn,^{1,2} Terry Quinn,¹ and Patrick B. Mark^{1,2}

¹Institute of Cardiovascular and Medical Sciences, University of Glasgow, Glasgow, UK; ²The Glasgow Renal & Transplant Unit, Queen Elizabeth University Hospital, Glasgow, UK; and ³Department of Neuroradiology, Queen Elizabeth University Hospital, Glasgow, UK

ABSTRACT

Background The immediate and longer-term effects of hemodialysis on cerebral circulation, cerebral structure, and cognitive function are poorly understood.

Methods In a prospective observational cohort study of 97 adults (median age 59 years) receiving chronic hemodialysis, we used transcranial Doppler ultrasound to measure cerebral arterial mean flow velocity (MFV) throughout dialysis. Using a well validated neuropsychological protocol, we assessed cognitive function during and off dialysis and after 12 months of treatment. We also used brain magnetic resonance imaging (MRI) to assess atrophy, white matter hyperintensities (WMHs), and diffusion parameters, and tested correlations between MFV, cognitive scores, and changes on MRI.

Results MFV declined significantly during dialysis, correlating with ultrafiltrate volumes. Percentage of decline in MFV correlated with intradialytic decline in cognitive function, including global function, executive function, and verbal fluency. At follow-up, 73 patients were available for repeat testing, 34 of whom underwent repeat MRI. In a subgroup of patients followed for 12 months of continued dialysis, percentage of decline in MFV correlated significantly with lower global and executive function and with progression of WMH burden (a marker of small vessel disease). Twelve of 15 patients who received renal transplants during follow-up had both early and follow-up off-dialysis assessments. After transplant, patients' memory (on a delayed recall test) improved significantly; increased fractional anisotropy of white matter (a measure of cerebral diffusion) in these patients correlated with improving executive function.

Conclusions Patients undergoing hemodialysis experience transient decline in cerebral blood flow, correlating with intradialytic cognitive dysfunction. Progressive cerebrovascular disease occurred in those continuing dialysis, but not in transplanted patients. Cognitive function and cerebral diffusion improved after transplant.

J Am Soc Nephrol 30: 147-158, 2019. doi: https://doi.org/10.1681/ASN.2018050462

Cognitive impairment is highly prevalent in those on hemodialysis (HD); up to 70% in an unselected cohort demonstrated cognitive impairment.¹ The etiology of cognitive impairment in HD is multifactorial: dialysis is associated with profound metabolic abnormalities, chronic inflammation, and oxidative and hemodynamic stressors. However, two important findings support the theory that cerebrovascular disease is the primary driving force. Specifically, dialysis preferentially induces deficits in executive function,^{2–4} a pattern traditionally observed early in vascular cognitive impairment but only a later complication of other dementias such as Alzheimer disease,⁵ and that more regular frequent dialysis does not improve cognitive function,⁶ whereas renal transplantation can.⁷

Received May 2, 2018. Accepted October 8, 2018.

Published online ahead of print. Publication date available at www.jasn.org.

Correspondence: Dr. Mark Findlay, Institute of Cardiovascular and Medical Sciences, University of Glasgow, 126 University Place, G12 8TA Glasgow, UK. Email: mark.findlay@nhs.net

Copyright © 2019 by the American Society of Nephrology

The frequency of cerebrovascular disease in people with ESRD is up to ten-fold greater⁸ compared with the general population and mortality rates approximately three-fold greater.⁹ Although conventional stroke risk factors such as hypertension are ubiquitous in ESRD and partly responsible, it is likely renal-specific factors contribute to this inordinate difference. For example, there is increasing evidence that both initiation¹⁰ of, and exposure^{11,12} to HD for ESRD is associated with stroke, although the mechanism is not fully explained. Interestingly, this risk appears to be reduced by receiving a renal transplant.¹³

The concept that HD can induce cerebral injury is supported by previous studies. Two recent prospective European studies have demonstrated evidence of decreased intradialytic cerebral perfusion,^{14,15} relating this to dialysis-related factors such as intradialytic hypotension or ultrafiltration volume. Alterations in cerebral structure are also described. For example, white matter hyperintensities (WMHs), a marker of small vessel disease, are present incidentally in 11%–21% of the general population aged around 64 years,¹⁶ whereas this rises to 52% in those on HD.¹⁷ Increasing WMH burden is associated not only with stroke,¹⁸ but also cognitive impairment.¹

Real-time, noninvasive measurement of cerebral blood flow is possible using transcranial Doppler (TCD) ultrasound, a noninvasive, inexpensive tool that is well tolerated and capable of performing repeated measurement throughout a dialysis session. Our primary hypothesis is that cerebral blood flow, measured by TCD, is reduced by HD and correlates with a decline in intradialytic cognitive function. Second, we hypothesize that a reduction in cerebral blood flow is capable of reducing longer term cognitive function, inducing ischemic damage as measured by WMH burden and that progressive ischemia on brain magnetic resonance imaging (MRI) at 12 months will correlate with progressive cognitive decline.

Therefore, we present a prospective investigation of realtime dialysis related cerebral blood flow, cognitive function on and off dialysis and brain MRI findings over a 12-month follow-up of continuous treatment for ESRD.

The primary aim of this study was to explore whether HD was associated with changes in cerebral blood flow and determine whether these changes relate to intradialytic cognitive dysfunction. Our secondary aims focus on (1) assessment of longitudinal changes in cerebral structure and function using brain MRI and cognitive assessment at baseline and after 12 months, correlating this to alterations in cerebral blood flow; and (2) to compare cognitive outcomes at follow-up in those who remain on HD and those who receive a renal transplant.

METHODS

Recruitment began in April 2015 and the first visit took place on June 3, 2015. All visits were completed by July 7, 2017.

Significance Statement

Recent data suggest that hemodialysis can reduce cerebral blood flow, and many assume—given the reported propensity of dialysisrelated cognitive impairment to present as executive dysfunction that alterations in cerebral circulation are implicated in cognitive decline. In this prospective observational study, using real-time vascular imaging (to measure cerebral blood flow) and neurocognitive assessments during a single dialysis session and after 12 months of ongoing treatment, the authors demonstrate a correlation between dialysis-related decrease in cerebral blood flow and measures of executive function. Brain imaging showed an increase in markers of small vessel disease in those remaining on dialysis. In transplanted patients, cerebral diffusion (and cognitive function) improved after transplant. These findings point to reduced cerebral blood flow as a mechanism of cerebral injury in hemodialysis.

Participants were asked to attend three visits over a 12-month follow-up period. Two visits took place within the first month: one during their routine dialysis treatment (intradialytic assessment), and a second on a nondialysis day (interdialytic assessment). A repeat nondialysis assessment was performed at 12 months.

Study Participants

All adult patients aged 18–85 years inclusive, receiving hospital HD for ESRD at the Glasgow Renal and Transplant unit were considered. We approached all adults who were expected to remain on hospital HD for \geq 6 months. Exclusion criteria were inability to consent, poor comprehension of English language, a prior diagnosis or neuroimaging evidence of cerebrovascular disease, and prior diagnoses of cognitive impairment. Neuroimaging evidence of cerebrovascular disease included all those with prior ischemic or hemorrhagic stroke, in addition to those with evidence of small vessel disease. Written informed consent was completed for each participant. This study was approved after ethical review by the West of Scotland Research Ethics Committee 5 (reference REC 15/WS/0024), and registered on Clinicaltrials.gov under identifier NCT02393222.

Clinical Variables

Patient demographics were acquired from the electronic patient record. Duration of ESRD was calculated in years from date of first RRT commencement until date of first visit. Preand post-BP and ultrafiltration volumes were recorded and presented as an average of the last six readings. Laboratory values utilized the three latest results preceding their first visit. During the intradialytic assessment visit, BP and weight were recorded before and after dialysis, and ultrafiltration rate was calculated as volume of fluid removed divided by duration of session.

Dialysis Schedule

This observational study was designed to assess the effect of dialysis in a "real-world" cohort, without intervention or

interference with each participant's routine treatment schedule. All participants underwent their routine prescribed dialysis treatment. The median duration of dialysis was 4 (interquartile range [IQR], 4–4.5) hours, with median blood flow rates of 300 (IQR, 287.5–305.0) ml/min. Dialysate temperature throughout all units was set at 36.5°C. The first dialysis session after the "long gap" was completely avoided. All Doppler readings were performed with patients sitting upright. It was essential that patients remained comfortable during their cognitive assessments, therefore food and rest periods were permitted.

Cognitive Assessments

Assessments were carried out by medical and nursing staff after training in appropriate use of the assessments by an experienced psychologist. At the midpoint of our study, retraining was performed to ensure consistency of assessment. Intradialytic testing was performed within the first 2 hours of dialysis, omitting the first 15 minutes for initial physiologic monitoring and routine nursing checks. All physical assessment as part of routine dialysis was performed manually, by usual dialysis nursing staff. The dialysis unit staff were aware to keep interruption to a minimum and cognitive testing was halted during scheduled BP checks. Where practically possible, to account for environmental effects on cognitive function, participants returned to the dialysis unit for their interdialytic assessment.

A modified 30-minute National Institute of Neurologic Disorders and Stroke-Canadian Stroke Network neuropsychological battery was used to assess cognitive function.¹⁹ This well validated protocol was chosen because of its feasibility, acceptability and increased sensitivity at detecting vascular cognitive impairment. It consists of verbal and written assessments of multiple cognitive domains. Specifically, we included the Montreal Cognitive Assessment (MOCA), using the accepted cut-off of <26 to define cognitive impairment. Verbal fluency was assessed using phonemic and semantic fluency, executive function using Trail-Making tests A and B (TMTA, TMTB) and the letter-digit substitution test (LDST) and auditory-verbal memory via the Hopkins Verbal Learning Test (HVLT). Finally, an assessment of mood was performed using the Center for Epidemiologic Studies Depression Scale. Further information on each assessment is described in Supplemental Table 1. The MOCA was used to quantify "cognitive impairment," therefore presented as an education adjusted score (addition of one for those with 12 years or less of education). All other scores are presented as raw, unadjusted data.

In an attempt to blunt possible learning effects, all visits were held a minimal 14 days apart. Further, we used versions 7.1, 7.2, and 7.3 of the MOCA and HVLT 1, 2, 5, and 6 at differing visits throughout the study.

In cases where writing was not possible (for example, a participant's inability or unwillingness to move their dominant arm during dialysis; n=9), the MOCA-blind method was used, adjusting the cut-off for cognitive impairment <18, and the verbal LDST was performed.

Imaging

After training from a research neuro-sonographer, all TCD ultrasound was performed by one operator (M.D.F.). Bilateral insonation of the middle cerebral arteries via trans-temporal windows was attempted in all patients during their intradialytic visits using the ST3 Transcranial Doppler Spencer Technologies TCD (Redmond, WA) with Power M-Mode 150. Measurements were performed with a sample of 6-9 mm and 2-MHz probes held in situ using a Marc600 head frame. Participants were asked to wear this throughout their dialysis session. Power began at 100 mW/cm² power and was reduced to the lowest possible setting for patient comfort. TCD recordings started at 50 mm of depth and adjusted to achieve the clearest possible signal. In participants with bilateral temporal windows, a median value was calculated for analyses. Recording were taken approximately 15 minutes before commencing dialysis, at set time intervals throughout dialysis and within 30 minutes of completion (Figure 1). Change in mean flow velocity (MFV) was calculated as the reading at 30 minutes after completion minus the reading 15 minutes before commencing dialysis.

Brain MRI was performed in a subset of patients using 3-T MRI Siemens research scanners (MAGNETOM Verio or PRISMA; Siemens Healthcare, Erlangen, Germany), allocated randomly as the first 40 participants that consented to the study willing and with no contraindications to MRI. All follow-up imaging was performed on their original scanner to allow direct comparison. T1, T2, fluid attenuated inversion recovery, and diffusion tensor imaging sequences were used to determine markers of atrophy (determined by change in volume of cortical gray matter, normal-appearing white matter, and supratentorial cerebrospinal fluid volumes), WMH burden, and cerebral diffusion, calculated as mean diffusivity (MD) and fractional anisotropy (FA).

Image Analyses

MRI results were analyzed using volumetric analyses software. The first step in automated extraction of WMH volumes was to estimate the white matter area in each participant using atlasbased segmentation. A probability map of white matter was previously created from 313 volunteers aged 18-96 years,²⁰ and registered to each participant using nonlinear (diffeomorphic) registration to provide an initial estimate of white matter in each individual.²¹ Hyperintense outliers were identified on T2 fluid-attenuated inversion recovery by transforming each voxel to a standard (z) score.²² Voxels with $z \ge 1.5$ and within the estimated white matter area were initially defined as WMH. Final WMH estimates were defined by threedimensional Gaussian smoothing to reduce noise and account for partial volumes around WMH edges. Automatic WMH estimates were visually checked and stroke infarcts masked by a trained image analyst following STRIVE guidelines.²³ Normal-appearing tissues, including cortical gray matter, subcortical gray matter, cerebral normal-appearing white matter, and supratentorial cerebrospinal fluid, were segmented using



Figure 1. Haemodialysis related decline in cerebral mean flow velocity The change in MFV during dialysis session (n=82), is presented as a median value and IQR (error bars). TCD recordings were taken prior, during, and after completion of dialysis, demonstrating a significant decline in MFV after dialysis; weighted GEE P<0.001.

population-specific tissue probability maps, within-patient T1 intensity data, and adjoining voxel data. Normal-appearing tissue segmentations were checked and edited in the same manner as WMH.

Measures of cerebral diffusion included FA and MD that were calculated from the eigenvalues (magnitude of water movement in each diffusion direction) obtained with diffusion MRI. The BrainSuite Diffusion Pipeline^{24,25} corrected geometric distortions in diffusion images and coregistered diffusion and structural images. FA and MD were calculated in regions of WMH and normal-appearing white matter.

Change in Modality of RRT

The primary aim of our study is to observe the effect of HD treatment on cerebral function and structure. However, we acknowledge that many patients suitable for recruitment may be listed for renal transplantation. Therefore, patients who remained on HD and those who received a renal transplant remained under follow-up. Death, patient request, or use of peritoneal dialysis resulted in withdrawal from the study. A flow chart outlining recruitment, follow-up, and testing is available in Supplemental Figure 1.

Statistical Analyses

Baseline demographic variables are presented as medians and compared using Mann–Whitney U or chi-squared tests, as appropriate.

Primary Analyses

Our primary aim was to describe alterations in cerebral blood flow and correlate this with intradialytic cognitive variation. Thus, MFV is presented as a median of MFV of all patients, demonstrating the change of MFV throughout the dialysis session. We applied a weighted generalized estimating equation (GEE) to model change in MFV, which had small amounts of missing data missing at random (Supplemental Table 2). Cognitive scores are paired data, comparing either on or off dialysis or again at 12 months follow-up, thus the Wilcoxon signed-rank test was used. Correlation between percentage of decline in MFV, dialysis-related variables, and change in cognitive score were performed using Spearman rank correlation. Multiple cognitive tests were correlated against change in MFV. To account for multiple comparison errors, we have provided both the Bonferroni and false discovery rate correction, highlighting significant values within each table. Further details are provided in Supplemental Tables 3 and 4.

Secondary Analyses

At follow-up the cohort was divided into those remaining on HD and those with a renal transplant for comparison. Cognitive assessments off dialysis were compared at 0 and 12 months using Wilcoxon signed-rank test, and correlation with baseline percentage of decline in MFV was assessed using Spearman rank correlation. Changes in MRI findings at 0 and 12 months are also paired and were compared using Wilcoxon signed-rank test and correlated with changes in cognitive assessment as above. Selected correlation plots are available in Supplemental Figures 2–5.

Statistical analyses were performed on SPSS version 24 (IBM, Armonk, NY) and the GEE was implemented using the "PROC GEE" statement in SAS version 9.4 (2002–2012; SAS Institute Inc.).

Power Calculations

To assess cognitive function, we calculated 73 patients would be required to show that a decline in HD, on the basis of published baseline ranges of a mean rescaled executive function of 8.7 (SD 2.8)³ and assuming a decline of 1 during follow up, with an SD of 3. To allow for a 25% dropout rate, we aimed to recruit 97 patients.

Characteristic	No Cognitive Impairment	Cognitive Impairment	All	P Value
Median age, yr [IQR]	58 [49–66]	58 [51–67]	58 [50.5–66.5]	0.61
Women, <i>n</i> [%]	20 [44.4]	15 [34.9]	35 [39.8]	0.36
Active on transplant waiting list, <i>n</i> [%]	20 [44.4]	17 [39.5]	37 [38.6]	0.64
Ethnicity, n [%]				0.96
European	43 [95.6]	41 [95.3]	84 [95.5]	
Asian	2 [4.4]	2 [4.7]	4 [4.5]	
Primary renal diagnosis, <i>n</i> [%]				0.36
Diabetes	8 [17.8]	10 [23.3]	18 [20.5]	
GN	8 [17.8]	11 [25.6]	19 [21.6]	
Interstitial	11 [24.4]	5 [11.6]	16 [18.2]	
Multisystem	7 [15.6]	10 [23.3]	17 [19.3]	
Other	11 [24.4]	7 [16.3]	18 [20.5]	
Past medical history, <i>n</i> [%]				
Hypertension	37 [82.2]	41 [95.2]	78 [88.6]	0.05
Diabetes mellitus	14 [31.1]	15 [34.9]	29 [33.0]	0.71
Ischemic heart disease	11 [24.4]	8 [18.6]	19 [21.6]	0.51
Congestive cardiac failure	3 [6.7]	6 [14.0]	9 [10.2]	0.26
Peripheral vascular disease	3 [6.7]	3 [7.0]	6 [6.8]	0.95
Atrial fibrillation	5 [11.1]	4 [9.3]	9 [10.2]	0.78
Depression	8 [17.8]	10 [23.3]	18 [20.5]	0.52
Duration of ESRD, yr [IQR]	1.39 [0.61–3.34]	1.92 [0.52–4.39]	1.62 [0.57–4.02]	0.84
Dialysis-related variables, median [IQR]				
Pre-SBP, mm Hg	133.5 [120.8–153.3]	148.8 [129.3–168.0]	143.5 [121.4–158.5]	0.02
Pre-DBP, mm Hg	70.8 [65.2–81.7]	74.8 [65.0–79.8]	73.1 [65.1–80.7]	0.61
Post-SBP, mm Hg	122.7 [112.4–143.8]	135.7 [118.2–158.0]	130.7 [113.4–148.5]	0.03
Post-DBP, mm Hg	68.0 [59.5–80.2]	71.2 [61.0–77.7]	69.3 [60.1–78.2]	0.59
UF volume, L	2.0 [1.3–2.4]	2.1 [1.4–2.9]	2.1 [1.4–2.6]	0.13
UF rate, ml/h	-442.1 [-240.0 to -575.0]	-500.0 [-350 to -622.2]	-484.4 [-261.9 to -484.4]	0.32
Dialysis access, n [%]				0.28
AV access	34 [75.6]	28 [65.1]	62 [70.5]	
Central venous catheter	11 [24.4]	15 [34.9]	26 [29.5]	
Median laboratory values [IQR]				
Serum-adjusted calcium, mmol/L	2.39 [2.29–2.46]	2.37 [2.27–2.45]	2.38 [2.28–2.46]	0.59
Serum phosphate, mmol/L	1.76 [1.42–2.03]	1.74 [1.49–2.04]	1.75 [1.47–2.04]	0.90
Hemoglobin, g/L	117.7 [106.7–122.7]	110.7 [101.0–120.33]	113 [102.2–122.0]	0.13
Serum albumin, mmol/L	33.0 [31.7–35.3]	33.3 [29.7–34.3]	33.2 [30.8–35.0]	0.94
PTH, nmol/L	68.6 [42.6–98.2]	53.3 [32.3–103.5]	61.6 [37.4–99.0]	0.31
Urea reduction ratio	71.5 [69–77]	73.5 [70.0–76.0]	72.8 [69.5–77.0]	0.58
Median years of education [IQR]	12 [11.0–13.5]	12 [11.0–14.0]	12 [11–14]	0.71

Table 1. Demographics of the study cohort, split by presence or absence of mild cognitive impairment using MOCA score<26 on baseline nondialysis day assessment (n=88)</td>

SBP, systolic BP; DBP, diastolic BP; UF, ultrafiltration; AV, arteriovenous; PTH, parathyroid hormone.

The primary analysis for the MRI study was to detect correlation between TCD and MRI findings for WMH lesion burden. To detect correlation (r=0.5), using a two-sided test at 5% significance with 80% power, the required sample size was 29. To allow for 25% dropout (death and renal transplant), a minimum of 40 patients were required.

RESULTS

Ninety seven patients were recruited, with a median age of 59 (IQR, 51–67) years; 40 (41.2%) were women and 32 (33%) were diabetic. Median duration of dialysis at first

visit was 1.76 (IQR, 0.6–4.0) years. Eighty eight patients completed both baseline intra- and interdialytic visits. Median duration between baseline visits was 27 (IQR, 22–34) days.

Cognitive Impairment at Baseline Off Dialysis

Cognitive impairment was present in 43 (48.9%) participants who were not known to have cognitive impairment, on the basis of MOCA score <26. Those with cognitive impairment were more likely to have evidence of systolic hypertension; predialysis systolic BP 148.8 versus 133.5 mm Hg (P=0.02). There were no differences in age or duration of ESRD Demographics are demonstrated in Table 1.

Table 2.	Clinical	variables	and their	r corre	elation	to	
percentag	je of de	cline in m	niddle cer	ebral a	artery	MFV	(n=82)

-		
Variable	Spearman Rho	P Value
Ultrafiltration volume, ml	0.512	< 0.001
Ultrafiltration rate, ml/h	0.493	< 0.001
Δ SBP, mm Hg (pre $-$ post)	0.196	0.08
Δ DBP, mm Hg (pre-post)	0.163	0.14
Δ Weight, kg (pre $-$ post)	0.463	< 0.001
Δ MAP, mm Hg (pre-post)	0.219	0.05
Diabetes mellitus	-0.304	0.005

 Δ Values are calculated as pre minus post values, therefore the greater the fall after dialysis the higher the value. A positive correlation denotes that as percentage of decline becomes more positive (*i.e.*, a greater decline in MFV), the associated variable becomes more positive. SBP, systolic BP; DBP, diastolic BP; MAP, mean arterial pressure.

HD and Cerebral Blood Flow

Insonation of one or both middle cerebral arteries was possible in 82 participants, but was not visible or was deemed unreliable in the remainder. Eighteen out of 82 participants were unable to continuously wear the headset throughout dialysis: in eight participants the headset was ill-fitting or uncomfortable, and a further ten participants requested removal within the first 2 hours, with six complaining of a headache that relieved immediately on removal. MFV declined after dialysis, remaining lower up to 30 minutes after completion of dialysis (median MFV reading 47.38–40.75 cm/s; P<0.001; Figure 1). The median percentage of decline was -10% (IQR, -21.9% to 0.0%). The percentage of

decline in MFV correlated with ultrafiltration volume and rate and change in weight (rho 0.512, 0.493, and 0.463, respectively; P<0.001), but not change in absolute BP values. Change in mean arterial pressure and presence of diabetes weakly correlated with percentage of decline in MFV (Table 2). Changes in MFV per participant, split by presence of cognitive impairment are presented in Supplemental Figures 6–8.

Cognitive Function: On- and Off-Dialysis Assessments

Compared with their assessments during an off-dialysis period, participants scored lower in assessments of executive function during dialysis (processing speed [LDST], 22 versus 24.5; P<0.001 and attention/task-switching [TMTB] time taken 88.5 versus 75 seconds; P<0.001; Table 3).

Significant correlations between the percentage of decline in MFV during dialysis and a decline in TMTA (rho -0.454; P < 0.001), TMTB (rho -0.323; P = 0.01), and phonemic fluency (rho 0.302; P = 0.01) during dialysis were detected. Further, worsening scores on global cognitive assessment (MOCA) during dialysis correlated with percentage of decline in MFV (rho 0.270; P = 0.02; Table 3).

Follow-Up

Median follow-up was 1.05 (IQR, 1.0–1.1) years. At follow-up, 15 participants had been transplanted, four withdrew, and six died (Supplemental Figure 1). Those who were transplanted were younger (median, 51 [IQR, 40–63] versus 60 [IQR, 52–67] years)

Table 3. Differences in cognitive scores during and off dialysis (*n*=88), and correlation between percentage of decline in MFV and cognitive scores

Assessment, <i>n</i> =88	Intradialytic Assessment	Interdialytic Assessment	P Value	Correlation with Percentage of Decline	P Value
				in MFV, Rho	
Global cognitive function					
MOCA	25.0 [22.0, 27.0]	25.0 [21.5, 27.0]	0.44	0.270	0.02 ^a
Verbal fluency					
Semantic	18.0 [14.5, 21.5]	18.0 [15.0, 22.0]	0.28	0.172	0.14
Phonemic	33 [26.0, 41.5]	34.0 [25.0, 43.5]	0.49	0.302	0.01ª
Executive function					
ТМТА	38.0 [26.5, 51.0]	35.5 [26.9, 50.0]	0.63	-0.454	<0.001 ^ª , ^b
ТМТВ	88.5 [62.0, 136.0]	75.0 [54.0, 112.0]	< 0.001	-0.323	0.01 ^a
LDST	22.0 [18.0, 27.0]	24.5 [20.0, 30.0]	< 0.001	-0.170	0.15
Auditory-verbal memory					
HVLT					
Total recall	22.0 [19.5, 25.5]	20.0 [17.0, 23.0]	< 0.001	0.089	0.45
Delayed recall	7.0 [5.0, 9.0]	6.0 [4.0, 9.0]	0.13	0.098	0.41
Retention	80.0 [60.0, 90.0]	80.0 [60.0, 100.0]	0.60	-0.046	0.70
Discrimination	10.0 [9.0, 11.0]	10.0 [9.0, 11.0]	0.90	0.057	0.63
Mood					
CES-D	8.0 [4.0, 17.0]	9.0 [4.0, 17.5]	0.80	0.097	0.41

Difference in cognitive score is calculated as nondialysis score minus dialysis score. Positive rho indicates that as percentage of decline increases (*i.e.*, a greater fall in MFV), the difference in cognitive scores becomes more positive. In the above tests a positive difference in cognitive score describes a lower cognitive score during dialysis, with the exception of TMTA, TMTB, and CES-D, where the inverse is true. CES-D, Center for Epidemiologic Studies Depression Scale; FDR, false detection rate.

^aCorrelations that remain significant after FDR.

^bCorrelations that remain significant after Bonferroni correction.

Assessment	Baseline Assessment	Follow-Up Assessment	P Value	Correlation with Percentage of Decline in MFV, Rho	P Value
Continued hemodialy	ysis, n=61				
MOCA	24.0 [21.0, 27.0]	26.0 [23.0, 28.0]	< 0.01	0.276	0.04
Semantic	19.0 [15.0, 21.0]	18.0 [15.0, 21.0]	0.45	0.201	0.15
Phonemic	35.0 [28.0, 44.0]	37.0 [27.0, 46.0]	0.06	0.150	0.28
LDST	24.0 [20.0, 31.0]	26.0 [20.0, 31.0]	0.87	-0.085	0.55
TMTA	34.0 [26.0, 47.0]	31.0 [26.0, 45.0]	0.09	-0.209	0.15
TMTB	71.0 [49.0, 99.0]	66.0 [49.0, 99.0]	0.90	-0.403	0.005ª
HVLT					
Total recall	20.0 [17.0, 23.0]	21.0 [17.0, 26.0]	0.28	0.098	0.48
Delayed recall	7.0 [4.0, 10.0]	7.0 [4.0, 9.0]	0.97	0.243	0.08
Retention	80.0 [62.5, 100.0]	80.0 [55.6, 90.9]	0.45	0.219	0.15
Discrimination	11.0 [9.0, 12.0]	11.0 [9.0, 12.0]	0.92	0.149	0.30
CES-D	10.0 [5.0, 18.0]	10.0 [4.0, 18.0]	0.59	0.139	0.32
Transplanted at follow	w-up, n=12				
MOCA	26.0 [24.5, 27.0]	26.0 [25.0, 28.5]	0.23	0.086	0.81
Semantic	18.0 [15.5, 26.0]	21.0 [20.5, 23.5]	0.42	0.609	0.06
Phonemic	33.5 [28.5, 46.0]	36.0 [28.5, 42.0]	0.96	0.758	0.01
LDST	31.0 [26.0, 35.0]	33.0 [26.0, 35.0]	0.43	0.383	0.28
TMTA	35.0 [26.0, 37.0]	34.5 [20.5, 41.0]	0.63	-0.500	0.17
TMTB	63.0 [51.0, 108.0]	58.0 [45.0, 82.0]	0.33	-0.317	0.41
HVLT					
Total recall	21.0 [17.5, 24.0]	22.5 [19.5, 27.5]	0.35	0.329	0.35
Delayed recall	6.5 [4.5, 8.0]	9.5 [7.0, 10.0]	0.02	-0.123	0.74
Retention	68.3 [59.8, 94.4]	100.0 [79.8, 105.0]	0.03	-0.358	0.31
Discrimination	10.0 [10.0, 11.0]	11.0 [10.5, 12.0]	0.03	0.273	0.45
CES-D	6.5 [1.0, 13.0]	2.5 [1.0, 8.0]	0.31	-0.334	0.35

Table 4. Differences in cognitive scores at baseline and follow-up (using interdialytic assessments), and correlation between percentage of decline in MFV and cognitive scores in those remaining on hemodialysis and those receiving a kidney transplant

Difference in cognitive score is calculated as baseline minus follow-up score. Therefore, a positive rho value indicates that as percentage of decline increases (*i.e.*, a greater fall in MFV), the difference in cognitive change becomes more positive. In the above tests a positive difference in cognitive score denotes deterioration at follow-up, with the exception of TMTA, TMTB, and CES-D, where the inverse is true. CES-D, Center for Epidemiologic Studies Depression Scale; FDR, false detection rate. ^aCorrelations that remain significant after FDR.

and had a shorter duration of ESRD (median, 0.6 [IQR, 0.2–1.6] versus 2.1 [IQR, 0.7–4.5] years; Supplemental Table 5). Paired 0 and 12 month interdialytic assessments were possible in 61 participants who remained on HD and 12 who received a kidney transplant. Baseline and follow-up MRI data were available in 24 participants who continued HD and ten who were transplanted.

Cognitive Function at Follow-Up: Continued HD versus Transplanted

At the 12-month follow-up, those who remained on dialysis had an improvement in MOCA score (26 versus 24; P < 0.01). No other assessment revealed significant differences. In contrast, those who underwent transplant had no change in their MOCA score (26 versus 26; P=0.23), but demonstrated an improvement in verbal memory (delayed recall, 9.5 versus 6.5; P=0.02; retention, 100% versus 68.3%; P=0.03; and discrimination, 11 versus 10; P=0.03; Table 4).

In those who remain on dialysis, percentage of decline in MFV was significantly correlated with worsening score in global cognitive function (MOCA) and executive function (TMTB) on follow-up (rho 0.276; P=0.04 and rho -0.403; P=0.005, respectively; Table 4).

In those who received a transplant, an observed percentage of decline in MFV correlated with worsening phonemic fluency at follow-up (rho 0.758; P=0.01; Table 4).

MRI-Derived Structural Changes

In those who continued on dialysis, progressive lobar atrophy was evident in frontal, parietal, and temporal lobes (P<0.05; Table 5), and WMH burden increased (from 2.96 to 3.36 ml; P=0.02; Figure 2). There were no changes in diffusion tensor imaging measures, FA, and MD (Table 5).

After kidney transplant, a similar pattern of lobar atrophy was noted. However, there was no progression in WMH burden and FA in white matter increased (from 0.28 to 0.29; P=0.02; Table 5).

Correlating Structural Changes with Cognitive Function

There was no correlation between percentage of decline in MFV assessed at baseline and MRI findings.

In those who continue dialysis, greater frontal atrophy correlates with worsening score in global cognitive and executive

	,	,	1
Assessment	Baseline Assessment	Follow-Up Assessment	P Value
Continued hemodialysis, n=2	24		
Median lobar volume, ml			
Frontal GM	160.3 [150.8 to 166.3]	155.2 [147.4 to 162.3]	0.02
Parietal GM	93.9 [89.1 to 102.6]	92.9 [85.4 to 99.2]	0.03
Temporal GM	116.8 [110.5 to 122.8]	111.3 [105.6 to 121.6]	0.009
Occipital GM	60.4 [55.2 to 65.9]	58.3 [54.8 to 63.6]	0.08
WMH volume, ml	2.96 [0.87 to 6.07]	3.36 [0.82 to 7.35]	0.02
DTI			
FA_WM	0.27 [0.02]	0.26 [0.02]	0.41
FA_WMH	0.26 [0.04]	0.26 [0.03]	0.64
MD_WM 10 ³ m ² /s	0.9 [0.1]	0.9 [0.1]	0.19
MD_WMH 10 ³ m ² /s	1.2 [0.01]	1.2 [0.1]	0.05
Transplanted at follow-up, n=	=10		
Median lobar volume, ml			
Frontal GM	166.9 [159.9 to 185.0]	162.2 [157.4 to 186.4]	0.03
Parietal GM	102.6 [98.1 to 114.8]	101.6 [97.8 to 112.2]	0.05
Temporal GM	115.0 [111.6 to 136.6]	113.8 [108 to 134.5]	0.009
Occipital GM	64.2 [60.1 to 75.6]	64.8 [59.7 to 73.9]	0.33
WMH volume, ml	1.02 [0.79 to 2.01]	1.17 [0.64 to 1.25]	0.24
DTI			
FA_WM	0.28 [0.01]	0.29 [0.01]	0.02
FA_WMH	0.25 [0.03]	0.26 [0.03]	0.07
MD_WM 10 ³ m ² /s	0.8 [0.01]	0.8 [0.02]	0.62
MD_WMH 10 ³ m ² /s	1.1 [0.1]	1.1 [0.1]	0.72

 Table 5.
 Differences in neuroimaging parameters at baseline and follow-up in those who remained on dialysis and those who received a kidney transplant

DTI data in Table 5 is mean [SD]. GM, gray matter; DTI, diffusion tensor imaging; FA_WM, fractional anisotropy in white matter; FA_WMH, fractional anisotropy in white matter hyperintensities; MD_WM, mean diffusivity in white matter; MD_WMH, mean diffusivity in white matter hyperintensities.

function (MOCA, rho 0.454; P=0.04; TMTB, rho 0.620; P=0.01; and auditory-verbal memory, rho 0.805; P=0.01). An increase in WMH burden correlates with verbal fluency and mood (rho 0.585; P=0.01 and rho -0.485; P=0.03, respectively; Table 6).

In who received a kidney transplant, an increase in WMH burden correlates with worsening executive function (TMTB, rho -0.81; *P*=0.02). In contrast, an increase in FA correlates with an improvement in executive function (LDST, rho -0.886; *P*=0.02). Surprisingly, worsening frontal atrophy correlates with improvement in executive function.

DISCUSSION

This prospective study of people with ESRD encompassing blood flow and functional and structural assessments of the brain has several important findings. First, we demonstrated that cerebral blood flow declines during dialysis, correlating with ultrafiltration volumes and with a measurable decline in executive function. At follow-up, a correlation was detected between reduced cerebral blood flow and executive function in those who remained on dialysis but not those who were transplanted. After transplant, verbal learning and memory improved. Finally, brain MRI revealed progressive WMH burden in those who remain on dialysis, but not those who were transplanted. In contrast, after transplantation increased FA of the white matter correlated with improved executive function.

We believe this study supports the hypothesis that HD may be associated with short-term "cerebral stunning," and demonstrates progressive injury in those who remain on dialysis, whereas there is a demonstrable improvement in memory and white matter integrity in those who are transplanted. Early recognition of those at risk may limit this cerebral injury, which appears potentially reversible by transplantation.

Cognitive Function in ESRD: Incidence and Consequences

We have demonstrated in a "low-risk" HD population, without recorded cerebrovascular disease or cognitive impairment, that mild cognitive impairment is present in 48.9% of the population. Once ESRD is established, it is estimated that up to 70% of those on HD have cognitive impairment.¹ Surprisingly, in that particular study only approximately 3% had documentation of cognitive impairment. We demonstrated intradialytic cerebral blood flow declines

with increasing ultrafiltration volumes Therefore, in ESRD, failure to recognize cognitive impairment could perpetuate future cognitive decline: in those with existing cognitive impairment, patient concordance with fluid restriction is poor, the corresponding increase in ultrafiltration volumes compromises cerebral circulation, cognition worsens, and concordance worsens further. Morbidity aside, cognitive dysfunction is also associated with all-cause mortality^{25,26} and dialysis withdrawal.²⁸

Dialysis and Cognitive Dysfunction

Multiple factors are reported to influence cognitive function in CKD. Conventional risk factors such as age, diabetes, hypertension, and dyslipidemia are abundant; however, unconventional "renal" factors, including treatments for CKD (*e.g.*, the historical effect of aluminum toxicity,²⁹ or more recently, the effect of dialysis itself), are implicated.^{2,30–32} In contrast to neurodegenerative cognitive disorders where memory is predominantly affected, the pattern of cognitive impairment most commonly described in those with CKD is loss of executive function,^{2–4} a phenotype associated with vascular cognitive impairment. Further, more intensive dialysis does not improve function.^{6,7} This has led to the conclusion that cognitive impairment is driven by cerebrovascular disease.



Figure 2. White matter hyperintensities progress in those on continued haemodialysis. The images on the left represent baseline MR images and on the right MR imaged at follow-up following 12 months of treatment. Additionally, a loss of cerebral volume can be appreciated as enlargement of lateral ventricles. This figure demonstrates visible progression of WMH on standard T2 fluid-attenuated inversion recovery sequences (top row) and after application of volumetric analyses software (bottom row).

Previous data on cognitive variation around the dialysis session are conflicting. Authors have described both improvements^{32,33} and worsening³¹ of postdialysis cognition. To our knowledge, this is the first study to correlate an acute change in cerebral blood flow with real-time alterations in cognitive function. This has several important implications. Immediately, clinicians should be aware that detailed clinical discussions should not be undertaken during HD as comprehension and recall maybe affected. Further, this must prompt additional research designed to limit the damage observed on MRI. One clinical trial has demonstrated that dialysate cooling can provide stability in white matter integrity at 1 year of treatment.³⁴ The beneficial cognitive effects of renal transplantation require further attention and the benefit of expediting transplantation as a treatment for early cognitive dysfunction should not be overlooked.

Transplantation and Cognitive Dysfunction

Renal transplantation remains the gold-standard treatment for ESRD.³⁵ Transplantation reduces the risk of cardiovascular disease,³⁶ increasing life expectancy, and improves quality of life.³⁷ In our study, we observed an improvement in cognitive function after transplantation, specifically memory, in agreement with previous studies,³⁸⁻⁴⁰ albeit in a small cohort of our patients. FA is a marker of white matter integrity and deficits in diffusion are recognized to precede WMH formation and are associated with worsening executive function.⁴¹ In those who received a renal transplant, FA improved and correlated with improved executive function, whereas there was no demonstrable improvement in anisotropic diffusion in those remaining on HD. Previous authors have demonstrated improvement in executive function after transplantation,³⁹ a finding that does not appear to be achievable by more frequent dialysis. In an analysis of participants in the Frequent Hemodialysis Network, an improvement in auditory-verbal memory and processing speed was observed after transplantation, but not after 12 months of frequent HD (6 days a week).⁷ This is supportive of a side effect produced by HD, ameliorated by transplantation.

Limitations

This prospective, observational cohort study has demonstrated important findings in a real-world ESRD cohort. However, we recognize the following limitations. As an observational study we cannot prove causation because of inability to account for Table 6.Correlations of changes in neuroimaging with change in cognitivetesting over 12 months, in those who remain on hemodialysis and those receivinga kidney transplant

Assessment	Correlation with Δ WMH	P Value	Correlation with ∆FA_WM	P Value
Continued hemodialys	sis, n=21			
MOCA	0.172	0.46	0.225	0.48
Semantic	-0.028	0.90	-0.175	0.59
Phonemic	0.585	0.01	-0.365	0.24
LDST	0.129	0.58	0.116	0.72
TMTA	0.328	0.16	-0.278	0.41
ТМТВ	-0.384	0.12	-0.103	0.78
HVLT				
Total recall	0.271	0.23	0.327	0.30
Delayed recall	0.268	0.24	0.160	0.62
Retention	0.172	0.46	0.056	0.86
Discrimination	0.296	0.19	-0.381	0.22
CES-D	-0.485	0.03	0.140	0.66
Transplanted at follow	-up, n=9			
MOCA	-0.092	0.81	0.319	0.54
Semantic	-0.301	0.43	-0.600	0.21
Phonemic	0.460	0.21	-0.667	0.15
LDST	-0.101	0.80	-0.886	0.02
TMTA	0.071	0.87	0.200	0.75
ТМТВ	-0.810	0.02	-0.600	0.29
HVLT				
Total recall	0.025	0.95	-0.232	0.66
Delayed recall	0.522	0.15	0.353	0.49
Retention	0.483	0.19	-0.143	0.79
Discrimination	-0.173	0.65	0.152	0.77
CES-D	0.460	0.21	-0.143	0.79

For all neuroimaging results, values show follow-up minus baseline, hence a reduction in value with time will produce a negative result. Difference in cognitive score is calculated as baseline minus follow-up score. Therefore, a positive rho indicates that a positive neuroimaging change relates to a positive cognitive score at follow-up. In the above tests, a positive follow-up cognitive score denotes deterioration, with the exception of TMTA, TMTB, and CES-D, where the inverse is true. FA_WM, Fractional anisotropy in white matter; CES-D, Center for Epidemiologic Studies Depression Scale.

unmeasured variables. For example, we do not have data on bicarbonate or blood viscosity levels, which may be of greater significance than ultrafiltration volume on MFV. We excluded those with any known evidence of cerebrovascular disease, which resulted in a younger cohort. This reduces generalizability and makes it likely we have underestimated the prevalence of cognitive impairment and the effect of ongoing treatments on more "fragile" brains. Although we acknowledge this as a limitation, we also believe it makes our findings all the more striking. No attempts were made to standardize dialysis regimes. Although this could mask findings acutely, we believe this provides a more realistic view of dialysis. Use of TCD ultrasound to measure cerebral blood flow has limitations, namely operator variability and absence of acoustic windows. All TCD readings were performed by one trained operator to reduce variation and we acknowledge TCD was not possible in 15 (15%) of our recruited cohort. Our findings are correlated to MFV changes over a single dialysis session, with follow-up spanning an entire year. Generalizing one session to an entire year may affect correlations at follow-up. As TCD measurements were taken at fixed time points, it not possible to comment on unrecorded episodes of acute decline in MFV. Finally, throughout this article we refer to cerebral blood flow. It is important to clarify that TCD measures cerebral blood flow velocity, an indirect measure of cerebral blood flow, provided the diameter of the insonated vessel does not change during measurements. It is not known whether HD alters cerebral vessel diameter.

Attempts were made to limit a learning effect. Namely, multiple versions of MOCA and HVLT were used, a minimum of 14 days between assessments was allowed, and visit order was reversed in a randomly selected 20% of the cohort. Despite our attempts we cannot completely exclude this learning effect. Frontal atrophy correlated with improved executive function in transplanted patients. Although frontal atrophy is a marker of aging and executive function can improve after transplantation,39 it is important to acknowledge our low number of transplant patients at follow-up, which could introduce both type 1 and 2 errors. In particular, this small number may lead to chance association between MRI findings and change in cognitive function. We did not detect correlation between MFV and MRI changes. Only 24 patients completed follow-up scans while on continued dialysis. Therefore, correlating

percentage of MFV decline with MRI findings is underpowered and may account for the lack of effect. Finally, and with particular reference to the young age of our cohort, we did not detect progressive cognitive dysfunction, but rather, improvement in MOCA score at 12 months in those on continued dialysis. It is likely that the follow-up time was too short to detect a measurable difference in cognitive function in this recruited cohort. In conclusion, our study highlights the high frequency of occult cognitive impairment, affecting 50% of our prevalent dialysis cohort without known cerebrovascular disease. HD is capable of inducing a transient decline in cerebral blood flow, correlating with intradialytic cognitive dysfunction. Evidence of progressive cerebrovascular disease is demonstrated in those who remain on dialysis, whereas after transplantation, improvement in cognitive function and cerebral anisotropic diffusion is suggestive of reversibility. Urgent interventions are required to limit cerebral stunning and prevent progressive cognitive decline.

ACKNOWLEDGMENTS

M.D.F., J.D., and P.B.M. had the original idea for the study. M.D.F., T.Q., and D.M. coordinated data collection. D.A.D. was responsible for producing outputs after magnetic resonance imaging analyses and contributed to the statistical analysis. M.D.F. analyzed the data and constructed the first draft. K.P.F. was responsible for reporting clinical findings on MRI. All authors contributed to the final draft.

M.D.F. is funded by a Kidney Research UK Training Fellowship, grant number TF9/2015 and is supported by a grant from Darlinda's Charity for Renal Research. T.Q. is supported by a joint Chief Scientist Office/Stroke Association Senior Clinical Lectureship, grant number TSA PPA 2015/01_CSO. D.A.D. is supported by a Stroke Association postdoctoral fellowship, grant number TSAPDF2017/01.

DISCLOSURES

None.

This article contains the following supplemental material online at http:// jasn.asnjournals.org/lookup/suppl/doi:10.1681/ASN.2018050462/-/ DCSupplemental.

SUPPLEMENTAL MATERIAL

Supplemental Figure 1. Flow chart demonstrating recruitment, follow-up and participant attendance to study visits and MRI. CeVD, cerebrovascular disease.

Supplemental Figure 2. Correlation plot of ultrafiltration and percentage of decline in MFV (Spearman rho -0.512; P < 0.001). More positive ultrafiltration volumes correlate with more negative percentage of decline in MFV.

Supplemental Figure 3. Correlation plot of Δ TMTB (early visits) and percentage of decline in MFV (Spearman rho -0.323; *P*=0.01). Δ Score is calculated as score during dialysis minus score off dialysis. Thus, for TMTB, if time taken to complete on dialysis is longer then the Δ score is positive. A more positive Δ TMTB denotes worsening of executive function during dialysis.

Supplemental Figure 4. Correlation plot of Δ TMTB (long-term) and percentage of decline in MFV in those on continued HD (Spearman rho -403; *P*=0.005). Δ Score is calculated as score at follow-up minus score at baseline. Thus, for TMTB, if time taken to complete is longer at follow-up, Δ score is more positive. A more positive Δ TMTB denotes a worsening executive function at follow-up.

Supplemental Figure 5. Correlation plot of Δ CES-D (long-term) and Δ WMH volume in those on continued HD (Spearman rho 485; P=0.03). Δ Score is calculated as score at follow-up minus score at baseline. Thus, for CES-D, if depression scoring is greater at follow-up, Δ score is more positive. A more positive Δ CES-D denotes a worsening in depressive symptoms at follow-up. CES-D, Center for Epidemiologic Studies Depression Scale.

Supplemental Figure 6. Trajectory of MFV during dialysis, in those without cognitive impairment.

Supplemental Figure 7. Trajectory of MFV during dialysis, in those with cognitive impairment.

Supplemental Figure 8. Trajectory of MFV during dialysis, in those with and without cognitive impairment, grouped.

Supplemental Table 1. Modified Hachinski protocol of cognitive assessments used both on and off dialysis. All assessments could be competed in <1 hour. For assessments completed during dialysis, all tests were completed within the first 2 hours of dialysis.

Supplemental Table 2. Missing at random (MAR)- and missing completely at random (MCAR)-based results for weighted GEE, with time as an ordinal variable.

Supplemental Table 3. Correlation of change in cognitive assessment and MFV as demonstrated in Table 3, with false detection rate and Bonferroni corrections. For comparison, FDR is calculated using an α of 0.1, where each *P*-value=rank/number of test multiplied by 0.1.

Supplemental Table 4. Correlation of change in cognitive assessment and MFVas demonstrated in Table 4, with false detection rate and Bonferroni corrections. For comparison, FDR is calculated using an α of 0.1, where each *P*-value=rank/number of test multiplied by 0.1.

Supplemental Table 5. Demographic of study cohort, split by those who remained on dialysis, or were transplanted. AV, arteriovenous; DBP, diastolic BP; PTH, parathyroid hormone; SBP, systolic BP; UF, ultrafiltration.

REFERENCES

- Murray AM, Tupper DE, Knopman DS, Gilbertson DT, Pederson SL, Li S, et al.: Cognitive impairment in hemodialysis patients is common. *Neurology* 67: 216–223, 2006
- Kurella Tamura M, Vittinghoff E, Hsu C-Y, Tam K, Seliger SL, Sozio S, et al.: CRIC Study Investigators: Loss of executive function after dialysis initiation in adults with chronic kidney disease. *Kidney Int* 91: 948–953, 2017
- Sarnak MJ, Tighiouart H, Scott TM, Lou KV, Sorensen EP, Giang LM, et al.: Frequency of and risk factors for poor cognitive performance in hemodialysis patients. *Neurology* 80: 471–480, 2013
- Weiner DE, Scott TM, Giang LM, Agganis BT, Sorensen EP, Tighiouart H, et al.: Cardiovascular disease and cognitive function in maintenance hemodialysis patients. *Am J Kidney Dis* 58: 773–781, 2011
- Román GC: Vascular dementia: Distinguishing characteristics, treatment, and prevention. J Am Geriatr Soc 51[Suppl Dementia]: S296– S304, 2003
- 6. Kurella Tamura M, Unruh ML, Nissenson AR, Larive B, Eggers PW, Gassman J, et al.: Frequent Hemodialysis Network (FHN) Trial Group: Effect of more frequent hemodialysis on cognitive function in the frequent hemodialysis network trials. Am J Kidney Dis 61: 228–237, 2013
- Dixon BS, VanBuren JM, Rodrigue JR, Lockridge RS, Lindsay R, Chan C, et al.: FHN study: Cognitive changes associated with switching to frequent nocturnal hemodialysis or renal transplantation. *BMC Nephrol* 17: 12, 2016
- Seliger SL, Gillen DL, Longstreth WT Jr., Kestenbaum B, Stehman-Breen CO: Elevated risk of stroke among patients with end-stage renal disease. *Kidney Int* 64: 603–609, 2003
- Power A, Chan K, Singh SK, Taube D, Duncan N: Appraising stroke risk in maintenance hemodialysis patients: A large single-center cohort study. Am J Kidney Dis 59: 249–257, 2012
- Murray AM, Seliger S, Lakshminarayan K, Herzog CA, Solid CA: Incidence of stroke before and after dialysis initiation in older patients. J Am Soc Nephrol 24: 1166–1173, 2013
- Toyoda K, Fujini K, Fujimi S, Kumai Y, Tsuchimochi H, Ibayashi S, et al.: Stroke in patients on maintenance hemodialysis: A 22-year singlecenter study. Am J Kidney Dis 45: 1058–1066, 2005

- Findlay M, MacIsaac R, MacLeod MJ, Metcalfe W, Traynor JP, Dawson J, et al.: Renal replacement modality and stroke risk in end-stage renal disease-a national registry study [published online ahead of print October 24, 2017]. Nephrol Dial Transplant 10.1093/ndt/gfx291
- Lentine KL, Rocca Rey LA, Kolli S, Bacchi G, Schnitzler MA, Abbott KC, et al.: Variations in the risk for cerebrovascular events after kidney transplant compared with experience on the waiting list and after graft failure. *Clin J Am Soc Nephrol* 3: 1090–1101, 2008
- MacEwen C, Sutherland S, Daly J, Pugh C, Tarassenko L: Relationship between hypotension and cerebral ischemia during hemodialysis. JAm Soc Nephrol 28: 2511–2520, 2017
- Polinder-Bos HA, García DV, Kuipers J, Elting JWJ, Aries MJH, Krijnen WP, et al.: Hemodialysis induces an acute decline in cerebral blood flow in elderly patients. J Am Soc Nephrol 29: 1317–1325, 2018
- Ylikoski A, Erkinjuntti T, Raininko R, Sarna S, Sulkava R, Tilvis R: White matter hyperintensities on MRI in the neurologically nondiseased elderly. Analysis of cohorts of consecutive subjects aged 55 to 85 years living at home. *Stroke* 26: 1171–1177, 1995
- Naganuma T, Takemoto Y, Shoji T, Shima H, Ishimura E, Okamura M, et al.: Factors associated with cerebral white matter hyperintensities in haemodialysis patients. *Nephrology (Carlton)* 17: 561–568, 2012
- Debette S, Markus HS: The clinical importance of white matter hyperintensities on brain magnetic resonance imaging: Systematic review and meta-analysis. BMJ 341: c3666, 2010
- Hachinski V, ladecola C, Petersen RC, et al.: National Institute Neurological Disorders and Canadian Stroke Network vascular cognitive impairment harmonization standards. Stroke 37: 2220–2241, 2006
- Tustison NJ, Cook PA, Klein A, Song G, Das SR, Duda JT, et al.: Largescale evaluation of ANTs and FreeSurfer cortical thickness measurements. *Neuroimage* 99: 166–179, 2014
- Avants BB, Epstein CL, Grossman M, Gee JC: Symmetric diffeomorphic image registration with cross-correlation: Evaluating automated labeling of elderly and neurodegenerative brain. *Med Image Anal* 12: 26–41, 2008
- 22. Freedman D, Pisani R, Purves R: *Statistics*, New York, W.W. Norton & Company Inc., 2007
- Wardlaw JM, Smith EE, Biessels GJ, Cordonnier C, Fazekas F, Frayne R, et al.: STandards for ReportIng Vascular changes on nEuroimaging (STRIVE v1): Neuroimaging standards for research into small vessel disease and its contribution to ageing and neurodegeneration. *Lancet Neurol* 12: 822–838, 2013
- 24. Shattuck DW, Leahy RM: BrainSuite: An automated cortical surface identification tool. *Med Image Anal* 6: 129–142, 2002
- Bhushan C, Haldar JP, Choi S, Joshi AA, Shattuck DW, Leahy RM: Coregistration and distortion correction of diffusion and anatomical images based on inverse contrast normalization. *Neuroimage* 115: 269–280, 2015

- Griva K, Stygall J, Hankins M, Davenport A, Harrison M, Newman SP: Cognitive impairment and 7-year mortality in dialysis patients. *Am J Kidney Dis* 56: 693–703, 2010
- Drew DA, Weiner DE, Tighiouart H, Scott T, Lou K, Kantor A, et al.: Cognitive function and all-cause mortality in maintenance hemodialysis patients. Am J Kidney Dis 65: 303–311, 2015
- Findlay MD, Donaldson K, Doyle A, Fox JG, Khan I, McDonald J, et al.: Scottish Renal Registry (SRR): Factors influencing withdrawal from dialysis: A national registry study. *Nephrol Dial Transplant* 31: 2041– 2048, 2016
- Alfrey AC, LeGendre GR, Kaehny WD: The dialysis encephalopathy syndrome. Possible aluminum intoxication. N Engl J Med 294: 184– 188, 1976
- Williams MA, Sklar AH, Burright RG, Donovick PJ: Temporal effects of dialysis on cognitive functioning in patients with ESRD. Am J Kidney Dis 43: 705–711, 2004
- Murray AM, Pederson SL, Tupper DE, Hochhalter AK, Miller WA, Li Q, et al.: Acute variation in cognitive function in hemodialysis patients: A cohort study with repeated measures. *Am J Kidney Dis* 50: 270–278, 2007
- Griva K, Newman SP, Harrison MJ, Hankins M, Davenport A, Hansraj S, et al.: Acute neuropsychological changes in hemodialysis and peritoneal dialysis patients. *Health Psychol* 22: 570–578, 2003
- Schneider SM, Malecki AK, Müller K, Schönfeld R, Girndt M, Mohr P, et al.: Effect of a single dialysis session on cognitive function in CKD5D patients: A prospective clinical study. Nephrol Dial Transplant 30: 1551–1559, 2015
- Eldehni MT, Odudu A, McIntyre CW: Randomized clinical trial of dialysate cooling and effects on brain white matter. JAm Soc Nephrol 26: 957–965, 2015
- Suthanthiran M, Strom TB: Renal transplantation. N Engl J Med 331: 365–376, 1994
- Dimény EM: Cardiovascular disease after renal transplantation. Kidney Int Suppl 61: 78–84, 2002
- Burra P, De Bona M: Quality of life following organ transplantation. Transpl Int 20: 397–409, 2007
- Griva K, Thompson D, Jayasena D, Davenport A, Harrison M, Newman SP: Cognitive functioning pre- to post-kidney transplantation–a prospective study. Nephrol Dial Transplant 21: 3275–3282, 2006
- Gupta A, Lepping RJ, Yu ASL, Perea RD, Honea RA, Johnson DK, et al.: Cognitive function and white matter changes associated with renal transplantation. Am J Nephrol 43: 50–57, 2016
- Harciarek M, Biedunkiewicz B, Lichodziejewska-Niemierko M, Dębska-Ślizień A, Rutkowski B: Continuous cognitive improvement 1 year following successful kidney transplant. *Kidney Int* 79: 1353–1360, 2011
- Grieve SM, Williams LM, Paul RH, Clark CR, Gordon E: Cognitive aging, executive function, and fractional anisotropy: A diffusion tensor MR imaging study. AJNR Am J Neuroradiol 28: 226–235, 2007